Author’s response to reviews

Title: Believability of messages about preventing breast cancer and heart disease through physical activity

Authors:

Tanya Berry (tanya.berry@ualberta.ca)

Kelvin Jones (kejones@ualberta.ca)

Kerry Courneya (kerry.courneya@ualberta.ca)

Kerry McGannon (kmcgannon@laurentian.ca)

Colleen Norris (cnorris@ualberta.ca)

Wendy Rodgers (wendy.rodgers@ualberta.ca)

John Spence (jc.spence@ualberta.ca)

Version: 1 Date: 24 Oct 2017

Author’s response to reviews:

The changes have been made as requested.

All e-mails have been added to the title page.

It has been specified that written consent was provided.

We have clarified that the funders had no role in the design of the study, data collection, analysis, interpretation of data, nor in writing the manuscript.